Author's response to reviews

Title: Clinical presentation, demographics and outcome of Tuberculosis (TB) in a low incidence area: a 4-year study in Geneva, Switzerland

Authors:

Omar Kherad (Omar.Kherad@hcuge.ch)  
Francois R Herrmann (Francois.Herrmann@hcuge.ch)  
Jean-Pierre Zellweger (zellwegerjp@swissonline.ch)  
Thierry Rochat (Thierry.Rochat@hcuge.ch)  
Jean-Paul Janssens (Jean-Paul.Janssens@hcuge.ch)

Version: 3 Date: 16 December 2009

Author's response to reviews:

Dear Editor,

we made all the formatting changes as requested.

Major Revision:

We provided a functioning link to reference 15 as requested and we removed the term "et al" from the references, quoting all the authors.

Minor Revision:

Title page and abstract have been modified as requested.

We also removed the authors' qualifications from the manuscript and retitled the "Methods" section as requested.

Thank you very much and regards,

Omar Kherad, MD